Author's response to reviews

Title: Changes of gluteus medius muscle in the adult patients with unilateral developmental dysplasia of the hip

Authors:

RuiYu Liu (liuryu@126.com)
XiaoDong Wen (wenxd_1986@163.com)
Zhiqin Zhiqin Tong (dy513@126.com)
KunZheng Wang (kunzheng1955@163.com)
ChunSheng Wang (wcs420@163.com)

Version: 2 Date: 5 January 2012

Author's response to reviews: see over
Dec. 5, 2011

Dear editor of the BMC Musculoskeletal Disorders:

I am submitting the manuscript “Changes of gluteus medius muscle in the adult patients with unilateral developmental dysplasia of the hip” by Kun Zheng Wang for consideration for publication in Journal of BMC Musculoskeletal Disorders. I confirmed that the manuscript has not been published or under consideration or publication elsewhere. This manuscript provided some useful information of gluteus medius muscle in patients with unilateral developmental dysplasia of the hip for hip soft tissue reconstruction. Further, this study has been approved by the institutional review board of the Second Hospital affiliated to Xian JiaoTong University. Correspondence concerning the manuscript should be to the author Kunzheng Wang. I look forward to learning your response to our submission.

Sincerely

Kun Zheng Wang, M.D.

E-mail :kunzheng1955@163.com

Department of Orthopaedic

Second Hospital affiliated to medical college

Xi’an Jiaotong University, Xi’an Shaanxi, 710004, P. R. China